Author's response to reviews

Title: Molecular Identification of Adenovirus Causing Respiratory Tract Infection in Pediatric Patients at the University of Malaya Medical Centre

Authors:

Juraina Abd-Jamil (juraina@um.edu.my)
Boon-Teong Teoh (boonlim82@yahoo.com)
Eddy H Hassan (eddyz116@gmail.com)
Nuruliza Roslan (reszac@yahoo.com)
Sazaly AbuBakar (sazaly@um.edu.my)

Version: 4 Date: 11 June 2010

Author's response to reviews:

10 June 2010

Nina Titmus
Scientific Editor
BMC-series Journals
BioMed Central
Floor 6, 236 Gray's Inn Road
London
WC1X 8HL

Dear Ms Nina Titmus,

MS: 1945798523166436

Attached please find manuscript 1945798523166436 – “Molecular Identification of Adenovirus Causing Respiratory Tract Infection in Pediatric Patients at the University of Malaya Medical Centre” which has been revised to accommodate all the Editor’s comments.

The item-by-item reply to the Editor’s concerns is attached.

Please do not hesitate to contact me shall you need any additional clarification and/or revision.

Sincerely yours,

Sazaly AbuBakar
Corresponding Author
Item-by-item reply to Editor

“Experimental research that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee....”

The study was conducted in compliance to all the requirement of the University Malaya Medical Ethics Committee. A statement to clearly indicate this point has been inserted into the revised manuscript. (Pg 5, line 16)

“Informed consent must also be documented.”

The UMMC Medical Ethics Committee does not require Informed Consent documentation for studies which does not directly involved procedures/protocols affecting patients' well being or confidentiality.

“We strongly encourage you to include an Acknowledgements section between the Authors?”

Agreed. Acknowledgements section has been added to acknowledge the contributions of the Department Of Medical Microbiology of the UMMC. Do note that the principal investigator is also the Head of the Department who oversee the UMMC Virology repository. (Pg 11, line 14)

“Please also include their source(s) of funding. Please also acknowledge anyone who contributed materials essential for the study...

Please list the source(s) of funding for the study, for each author, and for the manuscript preparation in the acknowledgements section. Authors must describe the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.”

The study does not receive any specific funding support. Materials used for the study was obtained with permission from the respective responsible entity. A statement to indicate this has been included. (Pg 11, lines 15-17)

“Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.”

No one is named in the Acknowledgement section to necessitate obtaining permission.